Author's response to reviews

Title: Correction: Reduction in BMI z-score and improvement in cardiometabolic risk factors in obese children and adolescents. The Oslo Adiposity Intervention Study - a hospital/public health nurse combined treatment.

Authors:

Magnhild LP Kolsgaard (Magnhild.L.P.Kolsgaard@ous-hf.no)
Geir Joner (geir.joner@medisin.uio.no)
Cathrine Brunborg (Cathrine.Brunborg@ous-hf.no)
Sigmund A Anderssen (Sigmund.Anderssen@nih.no)
Serena Tonstad (Serena.Tonstad@ous-hf.no)
Lene F Andersen (l.f.andersen@medisin.uio.no)

Version: 2  Date: 3 May 2012

Author's response to reviews:

Correction of our article Reduction in BMI z-score and improvement in cardiometabolic risk factors in obese children and adolescents. The Oslo Adiposity Intervention Study - a hospital/public health nurse combined treatment.